Author's response to reviews

Title: Mn bioavailability by polarized Caco-2 cells: comparison between Mn gluconate and Mn oxyprolinate.

Authors:

Chiara Foglieni (foglieni.chiara@hsr.it)
Mariangela Cavarelli (mariangela.cavarelli@hsr.it)
Mariarosaria Piscopiello (mariarosaria.piscopiello@hsr.it)
Alessandro Fulgenzi (alessandro.fulgenzi@unimi.it)
Maria Elena Ferrero (mariaelena.ferrero@unimi.it)

Version: 3 Date: 7 July 2011

Author's response to reviews:

Dear Editor,

I send you the manuscript which has been revised accordingly to the requests of the reviewers.

Since one of the two reviewers didn’t require changes to the manuscript, we now reply to the minor comments of the Dr. Elad Tako:

1) We didn’t modify the abstract, because the Results section is as short as possibly
2) The specific objectives of our study have been added in the Background section (Page 4) in bold
3) References non consistent are changed (Ref 44) or deleted (ex Ref 6, 7)

I hope that the new version of the manuscript is now suitable for publication.

My best regards

ME Ferrero